Author's response to reviews

Title: What stops children with a chronic illness accessing health care: a mixed methods study in children with Chronic Fatigue Syndrome/Myalgic Encephalomyelitis (CFS/ME)

Authors:

Carly M Webb (m0702395@sgul.ac.uk)
Simon M Collin (simon.collin@bristol.ac.uk)
Toity Deave (toity.deave@uwe.ac.uk)
Andrew Haig-Ferguson (andrew.haig-ferguson@nhs.net)
Amy Spatz (aspatz@sgul.ac.uk)
Esther Crawley (esther.crawley@bristol.ac.uk)

Version: 4 Date: 26 October 2011

Author's response to reviews:

Acronym of CFS/ME has been spelled out in full in both the title and in the abstract.